

Virulence and transmission vary between Usutu virus lineages in Culex pipiens

Maxime Prat, Mélanie Jeanneau, Ignace Rakotoarivony, Maxime Duhayon,

Yannick Simonin, Giovanni Savini, Pierrick Labbé, Haoues Alout

► To cite this version:

Maxime Prat, Mélanie Jeanneau, Ignace Rakotoarivony, Maxime Duhayon, Yannick Simonin, et al.. Virulence and transmission vary between Usutu virus lineages in Culex pipiens. PLoS Neglected Tropical Diseases, 2024, 18 (6), pp.e0012295. 10.1371/journal.pntd.0012295 . hal-04709994

HAL Id: hal-04709994 https://hal.science/hal-04709994v1

Submitted on 26 Sep 2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Citation: Prat M, Jeanneau M, Rakotoarivony I, Duhayon M, Simonin Y, Savini G, et al. (2024) Virulence and transmission vary between Usutu virus lineages in *Culex pipiens*. PLoS Negl Trop Dis 18(6): e0012295. https://doi.org/10.1371/journal. pntd.0012295

Editor: Mariangela Bonizzoni, Universita degli Studi di Pavia, ITALY

Received: January 29, 2024

Accepted: June 14, 2024

Published: June 27, 2024

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0012295

Copyright: © 2024 Prat et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data and scripts are available at https://zenodo.org/records/11093366? oken=eyJhbGci0iJIUzUxMilsImlhdCl6MTcxNDQ 4NDY2MiwiZXhwljoxNzl1MTQ4Nzk5fQ.eyJpZCl **RESEARCH ARTICLE**

Virulence and transmission vary between Usutu virus lineages in *Culex pipiens*

Maxime Prat^{1,2}, Mélanie Jeanneau², Ignace Rakotoarivony², Maxime Duhayon², Yannick Simonin³, Giovanni Savini⁴, Pierrick Labbé¹, Haoues Alout₁²*

1 Institut des Sciences de l'Evolution de Montpellier, Université de Montpellier-CNRS-IRD, Montpellier, France, 2 UMR ASTRE, Univ Montpellier, INRAE-CIRAD, Montpellier, France, 3 Pathogenesis and Control of Chronic Infections, Université de Montpellier-INSERM-EFS, Montpellier, France, 4 OIE Reference Centre for West Nile Disease, Istituto Zooprofilattico Sperimentale "G. Caporale", Teramo, Italy

* haoues.alout@inrae.fr; haoues.alout@cirad.fr

Abstract

Usutu virus (USUV) is a zoonotic arbovirus infecting mainly wild birds. It is transmitted by ornithophilic mosquitoes, mainly of the genus Culex from birds to birds and to several vertebrate dead-end hosts. Several USUV lineages, differing in their virulence have emerged in the last decades and now co-circulate in Europe, impacting human populations. However, their relative transmission and effects on their mosquito vectors is still not known. We thus compared the vector competence and survival of Culex pipiens mosquitoes experimentally infected with two distinct USUV lineages, EU2 and EU3, that are known to differ in their virulence and replication in vertebrate hosts. Infection rate was variable among blood feeding assays but variations between EU2 and EU3 lineages were consistent suggesting that Culex pipiens was equally susceptible to infection by both lineages. However, EU3 viral load increased with viral titer in the blood meal while EU2 viral load was high at all titers which suggest a greater replication of EU2 than EU3 in mosquito. While their relative transmission efficiencies are similar, at least at low blood meal titer, positive correlation between transmission and blood meal titer was observed for EU3 only. Contrary to published results in vertebrates, EU3 induced a higher mortality to mosquitoes (i.e. virulence) than EU2 whatever the blood meal titer. Therefore, we found evidence of lineage-specific differences in vectorial capacity and virulence to both the vector and vertebrate host which lead to balanced propagation of both viral lineages. These results highlight the need to decipher the interactions between vectors, vertebrate hosts, and the diversity of arbovirus lineages to fully understand transmission dynamics.

Author summary

Usutu virus is a zoonotic arbovirus that emerged in the last decades in Europe, impacting wild bird and human populations. Several co-circulating USUV lineages differ in their virulence to vertebrate hosts but their relative transmission and effects on their mosquito vectors is still not known. We compared the vector competence and survival of *Culex pipiens* mosquitoes experimentally infected with two distinct USUV lineages. We showed

6ljJiYjhhZThjLTE4NTgtNGQyZS1iZDBkLThIMTIIZ TRmYTgzMSIsImRhdGEiOnt9LCJyYW5kb20iOil0 YTgyNDYzYThmYmNjNzRhMTdjNjJk0TkzMGE4 NzVmNCJ9.tXk_mfDPknpzN_n0lBrg4DCMMqggEv JGmCdPB8QJ5i7TzLeAHX7nG_Swi-9F2 OfxMFuK9kM1Xj8SK9lv8A-UEq.

Funding: This work has been funded in part by the ANR PRC "ArchR" ANR-20-CE34-0007 (2021-2024) to PL and HA. PhD fellowship to MP has been funded by the ANR PRC "ArchR" ANR-20-CE34-0007 and the Animal Health department of INRAE. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

that their relative transmission efficiencies are similar but EU3 was more virulent to mosquitoes than EU2. In addition to published results, we revealed lineage-specific differences in vectorial capacity and virulence to both the vector and the vertebrate host suggesting that propagation of both viral lineages might be balanced. Our study provides new insights into the transmission of an emerging arbovirus and highlights the need to investigate the interactions between vectors, vertebrate hosts, and the diversity of arbovirus lineages to fully understand transmission dynamics.

Introduction

Usutu Virus (USUV) is a mosquito-borne flavivirus (family: *Flaviviridae*, genus: *flavivirus*) and is part of the Japanese encephalitis serocomplex with other phylogenetically related virus like Yellow Fever virus (YFV), Japanese Encephalitis virus (JEV) and West Nile virus (WNV). It was first identified in 1959 [1] along the Usutu River in Swaziland. Like WNV, USUV is maintained in a transmission cycle involving wild birds and ornithophilic mosquitoes, mainly of the genus *Culex* [2]. There have been recurrent outbreaks of USUV among birds with a few instances of spillover into humans and other mammals in several European countries since its emergence in 1996 in Italy [3,4]. These outbreaks were characterized by high mortality rates in passeriform and strigiform birds, with some species, such as blackbirds (*Turdus merula*), being particularly affected. Infections in other animals have been reported, including bats of the genus *Pipistrellus* [5], dogs [6,7], ruminants [8], wild boar [9], shrews and rodents [10], horses [11,12,13], and humans [2,4]. However, infections in these non-avian hosts are likely not transmissive (dead-end hosts) as for West Nile virus [14].

Flavivirus RNA genomes experience high mutation rates, which may facilitate rapid adaptation to new environments with potentially new hosts and new vectors [15]. As population sizes increase and the virus becomes more widespread, there is a greater likelihood of fitnessenhancing mutations furthering spread within and across continents. Such adaptation may directly and indirectly affect patterns of viral transmission, virulence, and host specificity. Phylogenetic studies have shown that USUV isolates can be divided into eight lineages: five European lineages (EU1 to EU5) and three African lineages (AF1 to AF3) [16]. Many of these lineages have been found co-circulating in Europe (i.e., EU1, EU2, EU3, EU5, AF2, and AF3). EU2 is the most detected lineage in mosquitoes, birds, and humans in central and southern Europe, while EU3 is the most prevalent in northern Europe [4,17]. Analyses of viral pathogenesis in mice and human cell culture revealed significant variation in virulence among-lineage: EU2 lineage appeared to be the most virulent, as it was more pathogenic to mice and replicated more efficiently and persistently in human neuronal cells [18] while AF2 and EU3 lineages appeared to be the least virulent [18,19].

Vectorial capacity is defined as the transmission potential of a vector population and depends on both ecological (vector density m, daily host biting rate a, daily survival rate p) and biological traits (vector competence c', Extrinsic Incubation Period or EIP n), as summarized in Eq 1 [20,21]:

$$C = \frac{ma_1 a_2 p^n c'}{-\ln p} \tag{1}$$

Eq 1. Vectorial capacity equation for multi-host transmission. Parameters are vector density *m*, daily host biting rate *a* (a_1 is the biting rate on one host; a_2 on another host), daily survival rate *p*, vector competence *c*', and Extrinsic Incubation Period *n*.

These traits are themselves affected by environmental factors (temperature, humidity, chemical exposure) modulating vector-borne transmission. Assessing vectorial capacity is critical for understanding transmission dynamics and, managing and controlling arbovirus transmission.

Vector competence defines the intrinsic ability of mosquitoes to transmit pathogens—from feeding on an infected host to transmission to a new host, thus including within-vector viral replication and dissemination up to the salivary glands. Competent mosquitoes will acquire virus from an infected host and later transmit the virus to other susceptible birds, or other animals, including humans. Entomological surveillances have reported USUV infections in several ornithophilic mosquitoes, mainly of the *Culex* genus but also in *Culiseta annulata*, *Anopheles maculipenis*, *Aedes spp*, *Mansonia spp*, and *Ochlerotatus spp* [3]. However, their role as vector for USUV has only been demonstrated in four *Culex* species: *Cx. neavi*, *Cx. pipiens* (biotypes *pipiens* and *molestus*), *Cx. quinquefasciatus*, and *Cx. torrentium* [22,23].

Environmental, genetic, symbiotic, and ecological factors may affect the likelihood of disease transmission by vectors, but genetic factors in the arboviruses themselves may also play a significant role. Given the significant pathogenic difference on vertebrate hosts among USUV lineages, it is important to consider whether similar difference is observed in how these lineages interact with and are transmitted by their vectors. We considered this question within the context of the virulence-transmission tradeoff hypothesis. The hypothesis posits that while highly replicating viruses may be transmitted more readily, they risk killing the host before transmission can occur. Thus, natural selection might favor viruses that balance high transmission and host mortality, resulting in intermediate levels of virulence [24-26]. Evidence of virulence-transmission tradeoffs has been suggested for rodent malaria [27], Schistosoma [28], and myxoma [27], although more data is needed to confirm their generality [28]. There is some evidence that mosquito-borne pathogens can reduce the longevity of infected mosquitoes [29] and, therefore, their vectorial capacity, which calls into question whether the virulence-transmission tradeoff is also important in the vector. Additionally, because the transmission cycle of arboviruses involves the infection of arthropod and vertebrate hosts, arboviruses may face an additional tradeoff if adaptation to one of the hosts negatively affects adaptation to the other [30].

Here, we investigated the relative vectorial capacity for two USUV lineages (EU2 and EU3), which differ significantly in their virulence in the vertebrate host. Throughout the 14 days of extrinsic incubation period [31], we measured daily mortality, viral load, and infection prevalence in diverse tissues of *Cx. pipiens* biotype *pipiens* mosquitoes (hereafter named *Cx. p. pipiens*) following infectious blood meals containing high or low viral titers to better characterize *Culex*-USUV interactions. We tested a 100-fold variation of viral titer reflecting the variation of replication of EU2 and EU3 linages in vertebrate host [18]. Our analyses allowed us to test for differences between lineages in transmission and mortality as well as the relationship between these traits within the mosquito and compared to patterns observed in the vertebrate host.

Materials and methods

Mosquito lines

Culex p. pipiens larvae were collected in September 2021 from ponds in a protected natural area of the Camargue at La Tour du Valat (https://tourduvalat.org/, coordinate: 43.509297, 4.668958) and maintained in the insectarium. PCR of the polymorphic Cq11 microsatellite [32] was used to confirm that mosquitoes belong to the biotype *pipiens*. The *C. p. pipiens* larvae and adults were raised in a secured insectarium (Baillarguet insectarium platform,

Montpellier) at $27\pm1^{\circ}$ C, $65\pm8\%$ relative humidity, and a 14h light/10h dark photoperiod. After egg-hatching, approximately 500 larvae per tray were randomly dispatched into three to five plastic trays containing 1 liter of tap water and fed with Tetramin fish granules. At emergence, adults were fed *ad libitum* with a 5% honey solution and maintained in 30x30x30 cm cages. The day before blood feeding, mosquitoes were starved by removing the honey solution.

Viral strains

Two USUV isolates were used, the first belonging to the Europe 2 lineage (EU2, Accession Number [AN]: MT784899) and the second to the EU3 lineage (AN: KX601690) [33]. They were isolated after three passages on VERO cells (ATCC CCL-81) and one on C6/36 cells (ATCC CRL-1660). All virus stocks were then amplified on Vero cells in MEM media, and the supernatants stored at -80°C until experimental infections. EU2 and EU3 virus stocks were titrated to $10^{7.2}$ TCID₅₀/mL and $10^{6.8}$ TCID₅₀/ml, respectively (Tissue Culture Infectious Dose 50), using the SpearmanKärber method [34].

Infection by membrane feeding assay

Infectious blood feedings were carried out with 5- to 8-day-old females who had not previously taken a blood meal. The day before the infectious blood meal, small cages (15x15x15 cm) containing approximately 200 females were placed in climatic chambers in the BSL3 laboratory (BioSecurity Level 3) with the same constant environmental conditions as the insectarium but no access to the honey solution. On the day of infection, 2 ml of infectious blood was prepared by mixing 1.5 ml of heparinized turkey blood, 0.5 mM ATP as a phagostimulant [35], and 500 μ l of MEM containing virus at a final titer of 10⁴ TCID₅₀/ml (low titer) or 10⁶ TCID₅₀/ml (high titer). Mosquitoes were allowed to feed on infectious blood using a Hemotek system at 37°C for 3 hours in darkness. Control females were treated in the same manner but fed with blood containing ATP and MEM but lacking virus. After blood feeding, 500µl of infectious blood was stored at -80° C, and mosquitoes were anesthetized using CO₂ to sort out the fully engorged females. They were kept individually for 14 days in Drosophila tubes containing a moistened foam and topped with a veil on which sat a cotton ball soaked in a 5% honey solution that was renewed every 2-3 days. At the end of this period, saliva was collected from live mosquitoes who were first anesthetized to remove the wings and legs and then placed with their proboscis inserted into 20 μ L tips filled with 10 μ L of MEM + 10% Fetal Bovine Serum (FBS). Saliva was collected for 30 minutes and stored at -80°C. Then, the thorax, head, and midgut of each female were dissected and stored separately at -80°C. Experimental infections were repeated three times for each USUV lineage and blood titer, resulting in 12 independent experimental infections.

Survival analysis

Every day over the 14-day course of infection, dead mosquito females were recorded daily and stored at -80°C. Dead and surviving females were analyzed for USUV infection (see method below). Note that females that died at day 1 were removed from the analysis to avoid virus detection from undigested or partially digested blood in the midgut.

Vector competence analysis

Total RNA was extracted using the KingFisher Flex automat and the NucleoMag RNA isolation kit (Macherey-Nagel) according to the manufacturer protocols and stored at -80°C. The detection and quantification of USUV viral genome was carried out by RT-qPCR as described in Nikolay et al. [36] using the NEB Luna Universal One-step RT-qPCR kit and the primer set: USUV-Forward (5'-AAAAATGTACGCGGATGACACA-3') and USUV-Reverse (5'-TTTGG CCTCGTTGTCAAGATC-3'). The samples were analyzed in duplicate concomitantly with a negative control and serial dilutions of USUV RNA extracted from USUV virus stock as standard to quantify individual viral load. The limit of detection has been defined as the Ct value of the last standard dilution with reproductible detection (i.e. all replicates for a dilution are positives, see S1 Fig). Vector competence was assessed based on several parameters: i) the prevalence of infection is the proportion of infected tissues among mosquitoes alive at 14 dpi (days post-infection), and ii) the viral load is the number of virus genome copies per mL of extracted RNA in infected tissues. For a fine characterization of vector competence, we calculated the transmission efficiency (TE: the proportion of mosquitoes with infected saliva among all blood-fed mosquitoes), the dissemination efficiency (DE: the proportion of mosquitoes with infected thoraces among all blood-fed mosquitoes), the infection efficiency (IE: the proportion of mosquitoes with infected midguts among all blood-fed mosquitoes), the dissemination rate (DR: the proportion of mosquitoes with infected thoraces among mosquitoes with infected midguts), and the transmission rate (TR: the proportion of mosquitoes with infected saliva among mosquitoes with infected thoraces). The infection status of dead mosquitoes was determined by the detection of virus in the whole body by RT-qPCR.

Statistical analysis

We analyzed the outcome of USUV infections in *Cx. p. pipiens* using two parameters as response variables: i) the infection prevalence and ii) the viral load. We examined the effects of diverse explanatory variables: "lineage" (categorial variable with two levels: EU2 and EU3), "blood meal titer" (a categorical variable with two levels: low and high), "true blood meal titer" (viral titer in blood after blood-feeding, numerical variable), "wing size" (numerical variable) and "replicate" (categorical variable with six levels: A, B, C, D, E, and F).

All statistical analyses were conducted using the R software 4.2.0. We computed generalized linear mixed effect models using the glmmTMB package [37] with i) a binomial error structure for analyzing the prevalence and ii) a negative binomial error structure for the viral load. Maximal models included the variables "lineage", "blood meal titer", "true blood meal titer", "wing size" and all their interactions. The "replicate" variable was used as a random variable to account for the nested data structure, i.e., the correlations between individuals from the same feeding experiment, and was either selected or removed based on the lowest AIC. Failed virus quantification in one tissue was specified as NA (Non Available) and was not included in the analysis. The significance of variables and selection of the minimal model was assessed using the ANOVA procedure within the package car [38], which performs type III hypothesis likelihood ratio tests (LRT). Mean parameter estimates were computed and post-hoc tests carried out to assess the differences between estimates with corrections for multiple comparisons using the package emmeans [39]. Kaplan Meier survival analyses were carried out and compared with that of the control group. The differences between survival distributions were estimated using the asymptotic Log-Rank Test [40], and the Hazard Ratio (HR defined as the ratio of the risk of death) was estimated from the *riskRegression* package [41].

The relative vectorial capacity of EU3 over EU2 has been calculated using the estimated vector competence (c') and survival (p) from this study keeping all other parameters equals (m, a and n). After simplification, the equation of the relative vectorial capacity is:

$$\frac{C_{EU3}}{C_{EU2}} = \left(\frac{p_{EU3}}{p_{EU2}}\right)^{n} \times \frac{\ln p_{EU2}}{\ln p_{EU3}} \times \frac{c'_{EU3}}{c'_{EU2}}$$
(2)

Eq 2. Equation of relative vectorial capacity. The equation was simplified based on fixed values for the density (m), the biting rate (a) and the extrinsic incubation period (n = 14 days).

Results

We examined the vector competence of *Cx. p. pipiens* for two USUV lineages (EU2 and EU3) provided in blood meals containing two distinct viral titers (referred as blood meal titers). We measured infection parameters and mortality after 14 days. Overall, 559 blood-fed females (188 with EU2, 236 with EU3, and 135 controls) from 3 replicates (*i.e.*, blood-feeding assays) per blood meal titer and USUV lineage were analyzed.

Prevalence of USUV infection in diverse tissues

We first investigated the USUV prevalence in mosquito to analyze the different parameters over all tissues and then analyze the different tissues separately to characterize the vector competence by calculating the infection, dissemination, and transmission efficiencies (IE, DE, and TE, respectively; see *Vector competence analysis* in *Material and Method*). Considering all tissues together (S1 Table), the prevalence of infection was a significantly influenced by the "bloodmeal titer" ($\chi^2 = 7.61$; *p*-value = 0.0058) and the "bloodmeal titer x lineage" interaction ($\chi^2 = 6.22$; *p*-value = 0.012). Prevalence was higher when mosquitoes fed on blood meal containing a higher virus titer than the one fed with lower virus titer for EU3 but not for EU2. In addition, infection prevalence was significantly influenced by the "bloodmeal titer x tissue" interaction ($\chi^2 = 10.54$; *p*-value = 0.0051) suggesting a different impact of bloodmeal titer in the different tissues. The minimal model revealed a significant effect of "Lineage x Bloodmeal titer x Wing size" interaction indicating an influence of mosquito size on infection that depends on USUV lineage and on the bloodmeal titer.

Focusing on the different tissues, the statistical analyses showed: i) For IE (midguts), none of the variable had a significant effect on prevalence for EU2 and EU3 (see Table 1 and Fig 1A) (low titer: $IE_{EU2} = 0.24 \pm 0.19$ and $IE_{EU3} = 0.46 \pm 0.25$, *p*-value > 0.05; high titer: $IE_{EU2} = 0.45 \pm 0.24$ and $IE_{EU3} = 0.61 \pm 0.23$, *p*-value > 0.05; Fig 1A). ii). For DE (thoraces), none of the variables had a significant effect (Table 1 and Fig 1B; low titer: $DE_{EU2} = 0.14 \pm 0.11$ and $DE_{EU3} = 0.20 \pm 0.14$, *p*-value > 0.05; high titer: $DE_{EU2} = 0.38 \pm 0.21$ and $DE_{EU3} = 0.52 \pm 0.22$, *p*-value > 0.05, Fig 1B). iii). For TE (saliva), the "*Lineage x Blood meal titer*" interaction was significant ($\chi^2 = 6.45$; *p*-value = 0.011, Table 1 and Fig 1C), as well as the "*Lineage*" effect ($\chi^2 = 4.83$, *p*-value = 0.027). Difference in TE was not significant between EU2 and EU3 at low titer contrary to high titer: $TE_{EU2} = 0.07 \pm 0.04$ and $TE_{EU3} = 0.02 \pm 0.02$, *p*-value = 0.11) but significant at high titer: ($TE_{EU2} = 0.10 \pm 0.05$ and $TE_{EU3} = 0.25 \pm 0.1$, *p*-value = 0.03).

Infection efficiency (IE)			Dissemination efficiency (DE)			Transmission efficiency (TE)			
Variable	χ ²	d.f.	p-value	χ^2	d.f.	p-value	χ^2	d.f.	p-value
Lineage	3.67	1	0.051	2.49	1	0.11	4.83	1	0.027
Bloodmeal titer	0.42	1	0.51	1.03	1	0.30	0.27	1	0.59
Interaction	0.19	1	0.65	0.08	1	0.77	6.45	1	0.011

Parameters selected in the minimal generalized linear mixed models are presented. X^2 is the Chi-square value of the likelihood ratio test and d.f. is the degree of freedom. The p-values are bolded when significant.



Fig 1. Prevalence of USUV infection in different *Cx. p. pipiens* **tissues.** The proportions of individual mosquitoes infected 14 dpi with EU2 (green points) and EU3 lineages (orange points), at low (left) and high (right) bloodmeal titer, are indicated for three body parts: A) midgut, B) thorax and C) saliva. Circles represent each replicate, with their diameter proportional to the corresponding sample size. Black bars represent the mean of the three replicates with overall sample sizes of 84 for EU2 and 109 for EU3 at low bloodmeal titer, and 76 for EU2 and 70 for EU3 at high bloodmeal titer. Only the significant differences are represented with stars (*: *p*-value < 0.05).

https://doi.org/10.1371/journal.pntd.0012295.g001

However, EU3 lineage present a significant dose-effect on TE (TE_{EU3} low vs. TE_{EU3} high, p-value = 0.004, Fig 1C).

Overall, the dissemination rate (DR, the proportion of infected thoraces among mosquitoes with infected midguts) and the transmission rate (TR, the proportion of mosquitoes with infected saliva among those with infected thoraces; Table 2 and Fig 2) were below 100%, indicating the presence of infection barriers in mosquitoes preventing USUV dissemination outside the midgut and in mosquito saliva. Neither USUV lineage nor blood meal titer significantly affected the DR (Fig 2A; low titer: $DR_{EU2} = 0.14 \pm 0.11$, $DR_{EU3} = 0.18 \pm 0.13$; *p*-value = 0.45; high titer: $DR_{EU2} = 0.37 \pm 0.2$ and $DR_{EU3} = 0.52 \pm 0.2$, *p*-value = 0.09; EU2 low vs high *p*-value = 0.31, EU3 low vs high *p*-value = 0.21). By contrast, the TR was significantly

Table 2.	Dissemination rate and	transmission	rates of USUV	lineages by	Cx. p. pipiens.
----------	------------------------	--------------	---------------	-------------	-----------------

Blood meal titer	L	ow	High		
USUV lineage	EU2	EU3	EU2	EU3	
N infected midguts	44	42	34	42	
N infected thorax	32	28	30	36	
N infected saliva	9	3	8	17	
Dissemination Rate (DR)	0.14 +/- 0.11	0.18 +/- 0.13	0.36 +/- 0.2	0.51 +/- 0.21	
Transmission Rate (TR)	0.06 +/- 0.04	0.02 +/- 0.016	0.09 +/- 0.04	0.23 +/- 0.09	

Estimates from general linear mixed models are presented with standard error.



Fig 2. Dissemination rate (DR) and transmission rate (TR) of EU2 and EU3 lineages in *Cx. p. pipiens.* The mean proportions of mosquitoes with infected thoraces among those with infected midguts (DR, panel A) and of individual mosquitoes with infected saliva among those with infected thoraces (TR, panel B) are indicated for EU2 (green points) and EU3 lineages (orange points), at low (left) and high (right) viral titer. Circles represent each replicate, with their diameter proportional to the corresponding sample size. Black bars represent the mean of the three replicates. Sample sizes are presented in Table 2. Only the significant differences are represented with stars (*: *p*-value < 0.05).

https://doi.org/10.1371/journal.pntd.0012295.g002

lower for USUV EU3 at low titer but not different between titers for EU2 or between EU2 and EU3 at high titer (Fig 2B; low titer: $TR_{EU2} = 0.06 \pm 0.04$ and $TR_{EU3} = 0.02 \pm 0.016$, *p*-value = 0.11; high titer: $TR_{EU2} = 0.09 \pm 0.04$ and $TR_{EU3} = 0.23 \pm 0.09$, *p*-value = 0.02; EU2 low vs high *p*-value = 0.61, EU3 low vs high *p*-value = 0.004).

Viral load in mosquito tissues

The analysis of infected females showed that viral load (VL, expressed in genome copies GE/ mL) was generally lower for EU3 than EU2, but only at low titer and not in all tissues (Fig 3): the "Lineage x Blood meal titer" interaction was significant for VL in midguts and thoraces (respectively $\chi^2 = 180.8$; *p*-value < 0.0001 and $\chi^2 = 139.68$; *p*-value < 0.0001; Table 3). Post hoc analyses revealed that EU3 viral load was significantly lower than EU2 at low titer but not at high titer in both the midgut and thorax (Fig 3A: low titer: $VL_{EU2} = 1.15 \times 10^6 \pm 5.16 \times 10^6 \text{ GE}/$ ml and VL_{EU3} = 2.79 ± 1.36 GE/ml, *p*-value < 0.0001; high titer: VL_{EU2} =: $2.64 \times 10^6 \pm 1.38 \times 10^6$ GE/ml, $VL_{EU3} = 2.88 \times 10^6 \pm 1.37 \times 10^6$ GE/ml, *p*-value = 0.99; EU2 low vs high *p*-value = 0.6, EU3 low vs high *p*-value < 0.0001; Fig 3B: low titer: VL_{EU2} = $8.62 \times 10^6 \pm 4.24 \times 10^6$ GE/ml and $VL_{EU3} = 33.4 \pm 11.7 \text{ GE/ml}, p$ -value < 0.0001; high titer: $VL_{EU2} = 3.07 \times 10^6 \pm 1.56 \times 10^6 \text{ GE/ml},$ $VL_{EU3} = 1.75 \times 10^{6} \pm 8.36 \times 10^{5} \text{ GE/ml}, p$ -value = 0.85; EU2 low vs high p-value = 0.46, EU3 low vs high *p*-value < 0.0001). In saliva, viral load was much lower than in the other tissues (Fig 3C) and was slightly higher in mosquitoes that had taken a blood meal with a higher titer (χ^2 = 5.88; p-value = 0.015). Post hoc analysis showed that there was only a slightly significant increase in EU2 viral load between titers (Fig 3C; low titer: $VL_{EU2} = 594.1 \pm 416$ GE/ml vs high titer: $VL_{EU2} =: 7.08 \times 10^3 \pm 5.26 \times 10^3$, p-value = 0.02). All other pairwise comparisons were not significant.

Effect of USUV on mosquito survival

Mortality was analyzed in two ways. First, we compared survivorship between mosquitoes fed on infectious blood (EU2 or EU3) and control mosquitoes fed on non-infectious blood (Fig 4 and Table 4). For each blood meal titer (low and high), the Kaplan-Meier survival curves



Fig 3. Viral load of USUV EU2 and EU3 in different *Cx. p. pipiens* tissues. Viral load (genome $copy/\mu L$) is represented for each USUV EU2 (green dots) and EU3 (orange dots) infected individuals, at low and high bloodmeal titers, in A) midguts, B) thorax and C) saliva. The black bar represents the mean. Only significant differences are represented with stars (*: *p*-value < 0.05; ***: *p*-value < 0.001).

https://doi.org/10.1371/journal.pntd.0012295.g003

showed a significant impact of feeding on infectious blood (*i.e.* being exposed to the virus) compared to feeding on non-infectious blood (Log-rank test: High: $\chi^2_{df=2} = 20.3$; p-value < 0.0001; Low: $\chi^2_{df=2} = 25.1$; p-value < 0.0001, Fig 4). This was true at both low and high titer for EU3 (*p*-value < 0.001 and *p*-value = 0.027, respectively, in comparisons with controls). However, EU2 lineage's impact on survival was only significant when fed a high blood meal titer (*p*-value = 0.115 and *p*-value < 0.0001 at low and high titer, respectively). Interestingly, there were no differences in survival between mosquitoes infected with the EU2 and EU3 lineages at high titer (*p*-value = 1), but at low titers, feeding on blood with EU3 lineage had a greater impact on survival than with EU2 (*p*-value = 0.035). Overall, the Hazard Ratios (HR) after feeding on infectious blood were HR_{low} = 7.13 and HR_{high} = 6.81 fold for EU3, and HR_{low} = 2.95 and HR_{high} = 5.8 fold for EU2 (Table 4).

		Midgut		Thorax			Saliva		
Variable	χ ²	d.f.	p-value	χ ²	d.f.	p-value	χ ²	d.f.	p-value
Lineage	0.014	1	0.9	0.65	1	0.42	3.10	1	0.07
Blood meal titer	1.47	1	0.22	2.13	1	0.14	5.88	1	0.015
Interaction	180.8	1	<0.0001	139.68	1	<0.0001	0.72	1	0.39

Table 3. S	tatistical analyses	of USUV EU2 and EU3	viral load in mosquito tissues.
------------	---------------------	---------------------	---------------------------------

Parameters selected in the minimal generalized linear mixed models are presented. X^2 is the Chi-square value of the likelihood ratio test and d.f. is the degree of freedom. The p-values are bolded when significant.



Fig 4. Effect of infectious blood feeding on *Cx. p. pipiens* **survival**. Survival of mosquito blood-fed with USUV EU2 and EU3 at low (A) and at high bloodmeal titer (B) were compared to survival of mosquito fed with non-infectious blood, to assess the combined effects of exposition to and infection by the viruses. All fully blood-fed females have been followed during 14 days and dead females were recorded daily. Pairwise Log-rank tests were used to compare survival between females that fed on infectious versus non-infectious bloodmeal and only significant differences were represented (*: *p*-value < 0.05; ***: *p*-value < 0.001).

https://doi.org/10.1371/journal.pntd.0012295.g004

In the second analysis, we separated USUV-positive and USUV-negative mosquito midguts for comparisons with controls (Fig 5 and Table 4). We used the infection data obtained on midguts because this is the first place of mosquito-arbovirus interactions that could lead to an effect on mosquito life history traits. By taking into account the infection status we can better test whether replicating virus within mosquitoes, and not just exposure to the virus, impacts survival. For this analysis, we did not include females dead at day one because of the risk of detecting USUV from the blood in their midgut (Sample size: $EU2_{Low} = 89 EU3_{Low} = 123 EU2_{High} = 82 EU3_{High} = 82$).

At low blood meal titer, the survival of USUV-negative individuals (*i.e.*, no detection EU2 or EU3 viral genome) did not differ from controls (Log-rank test: $\chi^2_{df=2} = 2.1 p$ -value = 0.3; Fig 5A). However, this difference was significant for USUV-positive mosquitoes (Log-rank test: $\chi^2_{df=2} = 7 p$ -value = 0.03; Fig 5B). In pairwise comparisons with controls, the survival of EU2-positive mosquitoes did not differ statistically from controls (EU2 *vs.* CG p-value = 1, HR_{EU2} = 1.89 Fig 5B). In contrast, for EU3-positive mosquitoes there was a significant reduction in survival (EU3 *vs.* CG *p*-value = 0.024, HR_{EU3} = 4.4).

Table 4. Hazard ratio of USUV lineages associated death with 95% confidence intervals.

		Blood meal titer			Blood meal titer		
		Low	High	Infection status	Low	High	
USUV lineages	EU2 N = 188	2.95 1.0-8.6	5.82 2.1–15.7	Negative N = 92	0.58 0.1-4.9	2.38 0.6-8.9	
				Positive N = 79	1.89 0.4–7.9	2.32 0.5–9.7	
	EU3 N = 236	7.13 2.8–18.2	6.81 2.5–18.1	Negative N = 121	1.95 0.6–6.4	4.65 1.4–15.2	
				Positive N = 84	4.40 1.3–14.4	5.05 1.6–15.5	

N represents the sample size.



Fig 5. Effect of USUV EU2 and EU3 infection on *Cx. p. pipiens* **survival.** All fully blood-fed females have been followed during 14 days. USUV detection has been performed on dead and surviving mosquitoes except females that died the first day after blood feeding in order to avoid virus detection from undigested blood. The upper panel shows Kaplan Meier survival curves of USUV negative *vs.* control individuals (A) and positive *vs.* control individuals (B) at low bloodmeal titer. The lower panel shows Kaplan Meier survival curves of USUV negative *vs.* control individuals (C) and positive *vs.* control individuals (D) at high bloodmeal titer. Pairwise Log-rank tests were used to compare survival between groups and only significant differences were represented with their p-value (*: p-value < 0.05; **: p-value < 0.01).

https://doi.org/10.1371/journal.pntd.0012295.g005

By contrast, at high blood meal titer, both USUV-negative and USUV-positive individuals showed reduced survival in comparisons with control mosquitoes (Log-rank test: $\chi^2_{df=2} = 7.4$ *p*-value = 0.02 and Log-rank test: $\chi^2_{df=2} = 9.7$ *p*-value = 0.008, respectively), suggesting that exposure to infectious blood (Fig 5C) and presence of replicating virus (Fig 5D) are both responsible of the observed effect. In pairwise comparisons with controls, taking a blood meal containing a high titer of EU3 lineage significantly reduced survival in both USUV-positive (EU3_{pos} vs. CG *p*-value = 0.0046, HR_{EU3} = 5.03) and negative (EU3_{neg} vs. CG *p*-value = 0.018, HR_{EU3} = 4.64) individuals. By contrast, pairwise comparisons found no significant differences in the survival of USUV-positive or negative individuals that had taken a blood meal with EU2 lineage (*p*-values > 0.05, respectively) despite the significant difference observed when taken as a whole (Fig 4B).

Lastly, we tested whether viral load in USUV-positive mosquitoes affected survival. Cox proportional hazard regression did not show any impact of viral load on survival for either lineage or blood meal titer (Table 5).

Discussion

Assessing the vectorial capacity for vector-borne diseases is critical to understanding arboviral emergence and characterizing their epidemiological cycles. *Cx. p. pipiens* is a known vector of

Lineage	Blood meal titer	χ ²	d.f.	p-value
EU2	Low	0.65	1	0.421
EU3	Low	0.92	1	0.337
EU2	High	0.11	1	0.739
EU2	High	2.00	1	0.157
	8	2.00	-	01107

Table 5. Statistical analysis of viral load on mosquito survival.

Cox proportional analyses were performed on infected individuals only. X² is the Chi-square value of the likelihood ratio test and d.f. is the degree of freedom. The p-values are bolded when significant.

https://doi.org/10.1371/journal.pntd.0012295.t005

USUV, a virus primarily of birds that has undergone recent phylogenetic and pathogenic divergence as it has spread throughout Europe. While there is known variation among the viral lineages in their virulence and pathogenicity, there has yet been no study of potential among-lineage variation in vector competence in addition to the effect on mosquito survival (as a proxy of virulence). Indeed, as for many arboviruses, vectorial capacity and more specifically vector competence is often tested with only one reference lineage. This neglect of viral diversity of virulence in and transmission from the vector host may profoundly limit our understanding of disease emergence and propagation. This study sought to determine whether these lineages vary in their interaction with the mosquito vector. To better characterize these interactions, we tested two distinct viral titers in blood meal provided to mosquitoes because it influences vector susceptibility to Usutu, West Nile and St Louis encephalitis viruses [22,42,43] as well as virulence [44].

Vector competence differs between USUV lineages and depends on the blood meal titer

We first examined vector competence by determining viral prevalence of the two lineages in different tissues of *Cx. p. pipiens* (Fig 1). While *Cx. p. pipiens* was competent for both USUV lineages, there was considerable among-replicate and among-individual variability. These differences likely arise from multiple factors, including physiological and immunological response heterogeneity and genetic differences in mosquitoes affecting blood feeding and infection. This variability highlights the need in future studies for adequate sample sizes and replication of artificial blood-feeding assays to generate reliable data on vector competence.

Infection of the mosquito vector proceeds from viral particles entering the midgut following an infectious blood meal, to dissemination into other tissues outside of the midgut, and finally into the saliva where virus can be transmitted when the mosquito takes another blood meal. This within-host viral propagation may be slowed or halted by various barriers to infection (reviewed by Franz et al. [45]). Beyond the physical/cellular barriers, the arthropod immune system may be able at each step to fight the viral infection through many interacting pathways, (e.g., RNA interference (RNAi), JAK-STAT, Toll and Imd pathways [46]). We observed a decrease in USUV prevalence from the midgut, to the thorax, and then the saliva, suggesting barriers do indeed prevent the entrance and propagation of viral particles within mosquito vectors. Our analysis of the infection efficiency and dissemination and transmission rates point to the importance of such barriers and, more interestingly, that the effects of these barriers depend on the viral lineage and the blood meal titer (Fig 2 and Table 2).

The infection efficiency reflects the ability of viral particles to infect the midgut after feeding on infectious blood. Midgut infection barriers prevent viral particles from entering the midgut epithelium due, for instance, to an absence of appropriate surface receptors, as described in three *Ae. Aegypti* colonies for a Dengue virus strain [47]. In our data about 50% of the mosquitoes that had taken an infectious blood meal had infected midguts, regardless of the differences in blood meal titer, suggesting that the midgut infection barrier is relatively easily overcome. The infection efficiency of the two lineages did not differ significantly (Table 1), suggesting a similar mosquito susceptibility to infection (Fig 1A). Difference in prevalence between blood meal titers was significant when all tissues were analyzed together, suggesting that increasing viral titer in the inoculum may facilitate infection.

From the midgut, viral particles must then move into other tissues. Midgut escape barriers may prevent such dissemination, perhaps through mechanisms acting in the basal lamina of the midgut epithelium [48]. Such barriers will reduce the dissemination rate from the midgut to tissues in the thorax. While overall there was a slight reduction in viral prevalence in mosquito thoraces, there was no effect of viral lineage or titer provided to blood-feeding mosquitoes on the dissemination efficiency, indicating that these lineages are equally efficient in their ability to infect other tissues within mosquito vector, independent of the blood meal titer.

Viral particles must then move into the salivary glands and saliva for transmission to occur when the mosquito vector takes a blood meal. Once again, escape barriers may affect this step in viral propagation by halting or slowing infection of the salivary gland cells or release into the saliva. Because we detected the presence of viral genome in saliva but not infectious particles, our data thus provided evidence of viral replication and release of viral genomes into the saliva allowing to characterize the salivary gland barrier but we cannot conclude on the infectiousness on the saliva. We found that differences in the transmission efficiency and transmission rate of the EU3 lineage depended on the viral titer in the blood meal contrary to EU2 (Figs 1C and 2B and Tables 1 and 2) suggesting that a high titer appears necessary to reach a higher transmission rate for the EU3 lineage but did not for EU2 lineage. This suggests a dosedependent transmission efficiency for EU3 but not for EU2 lineage: a higher viral load may increase the likelihood of dissemination across the salivary gland barrier for EU3 (Fig 3B and Table 3).

Together, these observations suggest that EU2 replicates more efficiently in mosquitoes than EU3 and that the lineage specific variation to overcome infection barriers and disseminate to the next tissue depend on the inoculating dose of viral particles. Arboviruses may face a proximate viral fitness (defined as the number of viral particles within one host) tradeoff if adaptation to one host environment reduces fitness in the other host [30] due to taxonomic divergence. Our data did not indicate such proximate trade-off as Clé et al. [18] described a higher and continuous EU2 replication in human neuronal cells up to six days, while EU3 replication decreased two days post-infection. If similar differences exist in mosquitoes, the less effective EU3 replication could limit within-mosquito viral production when the initial inoculum is low. The viral load in EU2-infected mosquitoes would not be affected by the blood meal titer due in part to its higher replication rate: at high blood meal titer, EU2 viral load may have reached a plateau after only a few days. Infection and replication at earlier and intermediate time points are necessary to test this hypothesis. The poor EU3 replication might also explain the low transmission rate observed at low titer (Fig 3), suggesting that a minimal number of viral particles is required to overcome the salivary gland barrier. This barrier appeared much more difficult to overcome, as both lineages saw a sheer drop in viral load in saliva. It is also worth considering that viral propagation into the salivary glands may occur later in time postinfection. Characterization of the key factors affecting the salivary gland barrier will focus on future genomic and transcriptomic differences between mosquitoes infected by the different lineages and their consequences for vector competence for USUV.

USUV lineages differ in virulence in vector and vertebrate hosts

We also investigated the virulence of the two USUV lineages in the *Cx. p. pipiens* vector. We first analyzed the effect on mosquito survival of taking an infectious blood meal, which combines the effects of viral exposure and potential infection (viral replication and propagation, Fig 4). To distinguish the impact of infection alone, we then analyzed survival taking into account the infection status of the dead and surviving mosquitoes (Fig 5). Indeed, between 29 to 58% of blood-fed mosquitoes developed an infection (Fig 1A), allowing for an effective comparison of USUV-positive and USUV-negative females (Fig 5).

Overall, it appears EU3 lineage is more virulent to its mosquito host than EU2. Indeed, survival analysis suggests that both exposure to and infection by EU3 affect vector survival. Interestingly, the higher virulence of the EU3 lineage appears little affected by the blood meal titer and despite a lower viral load in mosquitoes (Figs 3 and 4 and Table 4). However, when fed a low blood meal titer, simple exposure to EU3 is not enough to reduce survival (Fig 5A and Table 4). Such reduction in survival could have a great impact on transmission because the daily survival rate is the most influential parameter of the vectorial capacity equation (Eq 1) [49].

The mechanisms promoting EU3's increased virulence in mosquito are not known. It seems likely that it is a consequence of the production of a virulence factor that is present in EU3 but absent in EU2. Such an effect has been reported previously for Dengue, Japanese encephalitis and Zika viruses in *Ae. aegypti, Cx. pipiens* and *Ae. albopictus*, where the secretion of NS1 protein in host blood is associated with higher infection and suppression of mosquito immune responses (JAK-STAT and ROS pathways) [50–52] as well as virulence in mammalian hosts [53]. The presence of an EU3-specific NS1 could play a role in the observed differences in virulence. There are three non-synonymous substitutions (R36K, S57T, E139G) among 12 polymorphic sites in the coding region of NS1 between EU2 (MT784899) and EU3 (KX601690) but the functional consequences of these substitutions are not known. In addition, mutations in regulatory non-coding sequences could also be at play. More investigations are required to explore these issues.

An alternative hypothesis is that the reduction in host survival is due to the cost of mounting an immune response. In the frame of the life-history trait theory, trade-offs involving mosquito immune response may arise if the response against pathogens damages host tissues or takes energy and resources away from other fitness-promoting traits such as survival or reproduction. We did not directly measure the USUV-associated immune response in mosquitoes, so we cannot assess the importance of such trade-offs in mosquitoes. The mechanisms of the reduction in host survival, either virus-mediated through the production of virulence factors, host-mediated through the costs of immunity, or a combination of host and viral factors, will require further investigation.

Different replication, different transmission risks

Epidemiologically, the impact of USUV infection on host survival means less transmission because females will take fewer blood meals—a positive for public health in this context of a newly emerging virus. As the costs of USUV exposition and infection on mosquito vector survival appears different between lineages, it is also expected to differently impact the transmission of these various USUV lineages. To compare the combined effects of viral replication and virulence between the two lineages, we calculated the relative vectorial capacity (Eq 2) using the estimated daily survival and infection/transmission efficiencies (Table 6), the most influential parameters of the vectorial capacity. We hypothesized that EIP does not differ between USUV lineages but it would be interesting to test for variation of EIP between lineages in future studies. We found that EU3 does indeed appear less transmissible at low blood meal

Table 6.	USUV	EU3	over	EU2	relative	vectorial	capacit	y for	Culex p	oipiens.
----------	------	-----	------	-----	----------	-----------	---------	-------	---------	----------

	Blood n	Blood meal titer		
	Low	High		
Vector infection efficiency	0.0013	0.0207		
Daily survival rate	0.99	0.8316		
Relative vectorial capacity (EU3/EU2)	0.036	0.190		

https://doi.org/10.1371/journal.pntd.0012295.t006

titer than EU2 (ratio EU3/EU2 = 0.036), because daily survival rate is relatively reduced; at high blood meal titer, while mosquitoes showed a similar vector competence between EU2 and EU3, the latter remained the less transmissible lineage due to its higher impact on vector survival (ratio EU3/EU2 = 0.19).

From an evolutionary point of view, the reduction in transmission from increased virulence reduces ultimate viral fitness (defined as the number of new infected hosts), resulting in natural selection favoring genotypes that optimize the trade-off between transmission and virulence. This trade-off is generally thought to arise because increasing viral replication has the dual effect of increasing the probability of transmission but also increasing virulence [54]. We found no evidence of such a relationship between neither replication and virulence nor a between replication and transmission in mosquitoes. Indeed, EU3 lineage had lower replication rates, at least at low blood meal titer, but a greater virulence in mosquitoes, the opposite of the expectation from the virulence-transmission trade-off hypothesis. One hypothesis to explain this pattern would be that the immune response against EU3 is deleterious to mosquitoes, leading to earlier death after blood-feeding, but this needs to be confirmed with more data on the extrinsic incubation period and the blood-feeding behavior of mosquito vectors on real vertebrate hosts.

Contrary to what we observed in the mosquito vector, EU2 appears more virulent than EU3 in the vertebrate host [19]. Reduced lifespan of infected vertebrate host would reduce the probability of EU2 transmission to mosquitoes. Although EU2 replicates to a higher level in vertebrates, we showed that prevalence of infection in mosquito midgut were not significantly different between EU2 at high titer and EU3 at low titer. Therefore, while more mosquitoes could be infected by EU3 due to a longer infectious period of the vertebrate hosts, the increased virulence and lower replication rate in mosquito may reduce transmission of EU3 to a new vertebrate host. Therefore, considering the impact of USUV on survival of both vertebrate hosts and mosquito vectors, the propagation of the two lineages should be overall balanced, because of their opposite effect (Fig 6). These results suggest a trade-off between virulence, measured as reduction of survival, and transmission taking into account the whole transmission cycle with both hosts and vectors.

Conclusion

This study confirms that *Cx. p. pipiens* is a competent vector for USUV and that infection dynamics differ depending on the viral lineage and blood meal titer. At low blood meal titers, EU2 displayed higher viral load and was more often found in saliva, while EU3 is more virulent; the combined effect was that the relative vectorial capacity of mosquitoes infected with EU3 was much lower than with EU2. At high blood meal titers, viral load between the lineages was similar, but the high virulence of EU3 in mosquitoes meant a continued reduction in vectorial capacity. The absence of a relationship between viral load and virulence is inconsistent with the virulence/transmission trade-off. However, our data, coupled with those in a murine model, suggests a possible trade-off between virulence and transmission considering both



Fig 6. Schematic synthesis of USUV lineage specific transmission. The vertebrate (mouse) and arthropods (mosquitoes) hosts are represented on the arbovirus cycle. USUV replication level in hosts are indicated with the (+). The induced mortality rates of the hosts are proportional to the skull size. The transmission intensity is represented by the width of the arrows. The contrasting fitness between the different hosts of each USUV lineage suggests similar propagations of both lineages overall. This figure was created with BioRender.com.

https://doi.org/10.1371/journal.pntd.0012295.g006

mosquito vector and vertebrate host and demonstrate that viral lineages could be maintained in a balanced equilibrium (Fig 6). Further investigation is required to identify the genetic factors that could explain the lineage-specific differences in virulence and vector competence. Lastly, our study stresses that to fully understand the epidemic dynamics of emergent arboviruses, it is necessary to study more than one viral lineage as well as several isolates of each lineage and not focus solely on vertebrate hosts.

Supporting information

S1 Table. Statistical analyses of infection prevalence in mosquito 14 days post infection. (DOCX)

S1 Fig. Limit of detection (LOD) of USUV genome by RT-qPCR. Ct values obtained by RTqPCR (Nikolay et al. 2014 [36]) were plotted against the corresponding viral RNA titers after serial dilutions (orange for EU3 and blue for EU2). Errors bars represent the standard error of the mean. Viral stocks were titrated at $10^{7.2}$ and $10^{6.8}$ TCID₅₀/ml for EU2 and EU3, respectively by the method of SpearmanKärber method [34]. RNA was extracted from 150µl of virus stock using the NucleoMag RNA isolation kit (Macherey-Nagel) and eluted in 50µl before 10-fold serial dilution. RT-qPCR of serial dilutions and negative control were performed in duplicate on every plate (N = 8) containing RNA from mosquito tissues. We defined the limit of detection (*i.e.* individuals are considered infected or uninfected above or below this limit, resp.) as the Ct value of the last standard dilution with reproductible detection (*i.e.* all replicates for a dilution are positives), here 34 [95CI: 33.11; 34.96], which corresponded to the 10^{-7} dilution, i. e to 1 TCID₅₀/mL or 10^{-3} TCID₅₀/µL of virus stock. (TIFF)

Acknowledgments

All experiments on arthropods have been performed on the Baillarguet insectarium platform (https://doi.org/10.18167/infrastructure/00001), member of the National Infrastructure EMERG'IN and of the Vectopole Sud network (http://www.vectopole-sud.fr/). The Baillarguet insectarium platform is led by the joint units Intertryp (IRD, CIRAD) and ASTRE (CIRAD, INRAE). We thank Y. Simonin (UMR PCCEI) and N. Dheilly (UMR VIRO) for providing EU2 and EU3 virus isolates, F. Munoz for support in data visualization. We also thank the staff from La Tour du Valat for allowing us to collect *Culex pipiens* larvae. We thank Mylène Weill for helpful comments on this manuscript.

Author Contributions

Conceptualization: Haoues Alout.

Data curation: Maxime Prat, Haoues Alout.

Formal analysis: Maxime Prat, Haoues Alout.

Funding acquisition: Pierrick Labbé, Haoues Alout.

Investigation: Maxime Prat, Haoues Alout.

Methodology: Maxime Prat, Mélanie Jeanneau, Haoues Alout.

Project administration: Pierrick Labbé.

Resources: Maxime Prat, Ignace Rakotoarivony, Maxime Duhayon, Yannick Simonin, Giovanni Savini.

Supervision: Haoues Alout.

Visualization: Maxime Prat.

Writing - original draft: Maxime Prat, Haoues Alout.

Writing - review & editing: Maxime Prat, Yannick Simonin, Pierrick Labbé, Haoues Alout.

References

- Williams MC, Simpson DIH, Haddow AJ, Knight EM. The Isolation of West Nile Virus from Man and of Usutu Virus from the Bird-Biting Mosquito Mansonia Aurites (Theobald) in the Entebbe Area of Uganda. Ann Trop Med Parasitol. 1964; 58: 367–374. https://doi.org/10.1080/00034983.1964.11686258 PMID: 14212897
- Nikolay B, Diallo M, Boye CSB, Sall AA. Usutu Virus in Africa. Vector-Borne Zoonotic Dis. 2011; 11: 1417–1423. https://doi.org/10.1089/vbz.2011.0631 PMID: 21767160
- 3. Clé M, Beck C, Salinas S, Lecollinet S, Gutierrez S, Van de Perre P, et al. Usutu virus: A new threat? Epidemiol Infect. 2019; 147: e232. https://doi.org/10.1017/S0950268819001213 PMID: 31364580
- 4. Cadar D, Simonin Y. Human Usutu Virus Infections in Europe: A New Risk on Horizon? Viruses. 2022; 15: 77. https://doi.org/10.3390/v15010077 PMID: 36680117
- Cadar D, Becker N, Campos R de M, Börstler J, Jöst H, Schmidt-Chanasit J. Usutu virus in bats, Germany, 2013. Emerg Infect Dis. 2014; 20: 1771–1773. https://doi.org/10.3201/eid2010.140909 PMID: 25271769
- 6. Durand B, Haskouri H, Lowenski S, Vachiery N, Beck C, Lecollinet S. Seroprevalence of West Nile and Usutu viruses in military working horses and dogs, Morocco, 2012: dog as an alternative WNV sentinel

species? Epidemiol Infect. 2016; 144: 1857–1864. <u>https://doi.org/10.1017/S095026881600011X</u> PMID: 26838515

- Montagnaro S, Piantedosi D, Ciarcia R, Loponte R, Veneziano V, Fusco G, et al. Serological Evidence of Mosquito-Borne Flaviviruses Circulation in Hunting Dogs in Campania Region, Italy. Vector-Borne Zoonotic Dis. 2019; 19: 142–147. https://doi.org/10.1089/vbz.2018.2337 PMID: 30160624
- García-Bocanegra I, Paniagua J, Gutiérrez-Guzmán AV, Lecollinet S, Boadella M, Arenas-Montes A, et al. Spatio-temporal trends and risk factors affecting West Nile virus and related flavivirus exposure in Spanish wild ruminants. BMC Vet Res. 2016; 12: 249. https://doi.org/10.1186/s12917-016-0876-4 PMID: 27829427
- Escribano-Romero E, Lupulović D, Merino-Ramos T, Blázquez A-B, Lazić G, Lazić S, et al. West Nile virus serosurveillance in pigs, wild boars, and roe deer in Serbia. Vet Microbiol. 2015; 176: 365–369. https://doi.org/10.1016/j.vetmic.2015.02.005 PMID: 25724332
- Diagne MM, Ndione MHD, Di Paola N, Fall G, Bedekelabou AP, Sembène PM, et al. Usutu Virus Isolated from Rodents in Senegal. Viruses. 2019; 11: 181. <u>https://doi.org/10.3390/v11020181</u> PMID: 30795524
- Ben Hassine T, De Massis F, Calistri P, Savini G, BelHaj Mohamed B, Ranen A, et al. First detection of co-circulation of West Nile and Usutu viruses in equids in the south-west of Tunisia. Transbound Emerg Dis. 2014; 61: 385–389. https://doi.org/10.1111/tbed.12259 PMID: 25065813
- Bażanów B, Jansen van Vuren P, Szymański P, Stygar D, Frącka A, Twardoń J, et al. A Survey on West Nile and Usutu Viruses in Horses and Birds in Poland. Viruses. 2018; 10: 87. <u>https://doi.org/10. 3390/v10020087 PMID: 29462983</u>
- Gaibani P, Rossini G. An overview of Usutu virus. Microbes Infect. 2017; 19: 382–387. https://doi.org/ 10.1016/j.micinf.2017.05.003 PMID: 28602915
- 14. Bowen RA, Nemeth NM. Experimental infections with West Nile virus. Curr Opin Infect Dis. 2007; 20: 293–297. https://doi.org/10.1097/QCO.0b013e32816b5cad PMID: 17471040
- Holmes EC. The Evolutionary Genetics of Emerging Viruses. Annu Rev Ecol Evol Syst. 2009; 40: 353– 372. https://doi.org/10.1146/annurev.ecolsys.110308.120248
- Siljic M, Sehovic R, Jankovic M, Stamenkovic G, Loncar A, Todorovic M, et al. Evolutionary dynamics of Usutu virus: Worldwide dispersal patterns and transmission dynamics in Europe. Front Microbiol. 2023; 14. Available from: https://www.frontiersin.org/articles/10.3389/fmicb.2023.1145981. https://doi.org/10. 3389/fmicb.2023.1145981 PMID: 37032910
- Vilibic-Cavlek T, Petrovic T, Savic V, Barbic L, Tabain I, Stevanovic V, et al. Epidemiology of Usutu Virus: The European Scenario. Pathogens. 2020; 9: 699. <u>https://doi.org/10.3390/pathogens9090699</u> PMID: 32858963
- Clé M, Constant O, Barthelemy J, Desmetz C, Martin MF, Lapeyre L, et al. Differential neurovirulence of Usutu virus lineages in mice and neuronal cells. J Neuroinflammation. 2021; 18: 11. <u>https://doi.org/10. 1186/s12974-020-02060-4</u> PMID: 33407600
- Kuchinsky SC, Hawks SA, Mossel EC, Coutermarsh-Ott S, Duggal NK. Differential pathogenesis of Usutu virus isolates in mice. PLoS Negl Trop Dis. 2020; 14: e0008765. https://doi.org/10.1371/journal. pntd.0008765 PMID: 33044987
- 20. Macdonald G The analysis of the sporozoite rate. Trop Dis Bull. 1952; 49: 569–586.
- Brady OJ, Godfray HCJ, Tatem AJ, Gething PW, Cohen JM, McKenzie FE, et al. Vectorial capacity and vector control: reconsidering sensitivity to parameters for malaria elimination. Trans R Soc Trop Med Hyg. 2016; 110: 107–117. https://doi.org/10.1093/trstmh/trv113 PMID: 26822603
- Nikolay B, Diallo M, Faye O, Boye CS, Sall AA. Vector Competence of Culex neavei (Diptera: Culicidae) for Usutu Virus. Am J Trop Med Hyg. 2012; 86: 993–996. <u>https://doi.org/10.4269/ajtmh.2012.11-0509</u> PMID: 22665607
- Holicki CM, Scheuch DE, Ziegler U, Lettow J, Kampen H, Werner D, et al. German Culex pipiens biotype molestus and Culex torrentium are vector-competent for Usutu virus. Parasit Vectors. 2020; 13: 625. https://doi.org/10.1186/s13071-020-04532-1 PMID: 33380339
- Anderson RM, May RM. Coevolution of hosts and parasites. Parasitology. 1982; 85: 411–426. <u>https://doi.org/10.1017/s0031182000055360</u> PMID: 6755367
- Ewald PW. Host-Parasite Relations, Vectors, and the Evolution of Disease Severity. Annu Rev Ecol Syst. 1983; 14: 465–485. https://doi.org/10.1146/annurev.es.14.110183.002341
- 26. Frank SA. Models of Parasite Virulence. Q Rev Biol. 1996; 71: 37–78. https://doi.org/10.1086/419267 PMID: 8919665
- Dwyer G, Levin SA, Buttel L. A Simulation Model of the Population Dynamics and Evolution of Myxomatosis. Ecol Monogr. 1990; 60: 423–447. https://doi.org/10.2307/1943014

- Alizon S, van Baalen M. Transmission–virulence trade-offs in vector-borne diseases. Theor Popul Biol. 2008; 74: 6–15. https://doi.org/10.1016/j.tpb.2008.04.003 PMID: 18508101
- Alout H, Dabiré RK, Djogbénou LS, Abate L, Corbel V, Chandre F, et al. Interactive cost of Plasmodium infection and insecticide resistance in the malaria vector Anopheles gambiae. Sci Rep. 2016; 6: 29755. https://doi.org/10.1038/srep29755 PMID: 27432257
- Coffey LL, Vasilakis N, Brault AC, Powers AM, Tripet F, Weaver SC. Arbovirus evolution in vivo is constrained by host alternation. Proc Natl Acad Sci. 2008; 105: 6970–6975. https://doi.org/10.1073/pnas. 0712130105 PMID: 18458341
- Krambrich J, Bole-Feysot E, Höller P, Lundkvist Å, Hesson JC. Vector competence of Swedish Culex pipiens mosquitoes for Usutu virus. One Health. 2024; 18: 100707. https://doi.org/10.1016/j.onehlt. 2024.100707 PMID: 38500563
- Bahnck CM, Fonseca DM. Rapid assay to identify the two genetic forms of Culex (Culex) pipiens L. (Diptera: Culicidae) and hybrid populations. Am J Trop Med Hyg. 2006; 75: 251–255. PMID: <u>16896127</u>
- Lecollinet S, Blanchard Y, Manson C, Lowenski S, Laloy E, Quenault H, et al. Dual Emergence of Usutu Virus in Common Blackbirds, Eastern France, 2015. Emerg Infect Dis. 2016; 22: 2225. <u>https://doi.org/ 10.3201/eid2212.161272 PMID: 27869608</u>
- Kärber G. Beitrag zur kollektiven Behandlung pharmakologischer Reihenversuche. Naunyn-Schmiedebergs Arch Für Exp Pathol Pharmakol. 1931; 162: 480–483. https://doi.org/10.1007/BF01863914
- Galun R, Avi-Dor Y, Bar-Zeev M. Feeding Response in Aedes aegypti: Stimulation by Adenosine Triphosphate. Science. 1963; 142: 1674–1675. https://doi.org/10.1126/science.142.3600.1674 PMID: 17834375
- 36. Nikolay B, Weidmann M, Dupressoir A, Faye O, Boye CS, Diallo M, et al. Development of a Usutu virus specific real-time reverse transcription PCR assay based on sequenced strains from Africa and Europe. J Virol Methods. 2014; 197: 51–54. https://doi.org/10.1016/j.jviromet.2013.08.039 PMID: 24036076
- Brooks M E, Kristensen K, Benthem K J, van, Magnusson A, Berg C W, Nielsen A, et al. glmmTMB Balances Speed and Flexibility Among Packages for Zero-inflated Generalized Linear Mixed Modeling. R J. 2017; 9: 378. https://doi.org/10.32614/RJ-2017-066
- An R Companion to Applied Regression. In: SAGE Publications Inc [Internet]. 28 Sep 2023 [cited 5 Oct 2023]. Available from: <u>https://us.sagepub.com/en-us/nam/an-r-companion-to-applied-regression/</u> book246125.
- Lenth RV, Bolker B, Buerkner P, Giné-Vázquez I, Herve M, Jung M, et al. emmeans: Estimated Marginal Means, aka Least-Squares Means. 2023. Available from: <u>https://cran.r-project.org/web/packages/</u> emmeans/index.html.
- 40. Cox DR. Regression Models and Life-Tables. J R Stat Soc Ser B Methodol. 1972; 34: 187–202. <u>https://</u>doi.org/10.1111/j.2517-6161.1972.tb00899.x
- Gerds T, Ohlendorff J, Ozenne B. riskRegression: Risk Regression Models and Prediction Scores for Survival Analysis with Competing Risks. 2023. Available from: <u>https://cran.r-project.org/package=riskRegression</u>.
- 42. Richards SL, Mores CN, Lord CC, Tabachnick WJ. Impact of Extrinsic Incubation Temperature and Virus Exposure on Vector Competence of Culex pipiens quinquefasciatus Say (Diptera: Culicidae) for West Nile Virus. Vector Borne Zoonotic Dis Larchmt N. 2007; 7: 629–636. https://doi.org/10.1089/vbz. 2007.0101 PMID: 18021028
- Richards SL, Lord CC, Pesko K, Tabachnick WJ. Environmental and Biological Factors Influencing Culex pipiens quinquefasciatus Say (Diptera: Culicidae) Vector Competence for Saint Louis Encephalitis Virus. Am J Trop Med Hyg. 2009; 81: 264–272. PMID: 19635881
- 44. Benzarti E, Rivas J, Sarlet M, Franssen M, Desmecht D, Schmidt-Chanasit J, et al. Experimental Usutu Virus Infection in Domestic Canaries Serinus canaria. Viruses. 2020; 12: 164. <u>https://doi.org/10.3390/ v12020164</u> PMID: 32023880
- Franz AWE, Kantor AM, Passarelli AL, Clem RJ. Tissue Barriers to Arbovirus Infection in Mosquitoes. Viruses. 2015; 7: 3741–3767. https://doi.org/10.3390/v7072795 PMID: 26184281
- Lee W-S, Webster JA, Madzokere ET, Stephenson EB, Herrero LJ. Mosquito antiviral defense mechanisms: a delicate balance between innate immunity and persistent viral infection. Parasit Vectors. 2019; 12: 165. https://doi.org/10.1186/s13071-019-3433-8 PMID: 30975197
- Mercado-Curiel RF, Black WC, Muñoz M de L. A dengue receptor as possible genetic marker of vector competence in Aedes aegypti. BMC Microbiol. 2008; 8: 118. <u>https://doi.org/10.1186/1471-2180-8-118</u> PMID: 18625079
- Carpenter A, Clem RJ. Factors Affecting Arbovirus Midgut Escape in Mosquitoes. Pathogens. 2023; 12: 220. https://doi.org/10.3390/pathogens12020220 PMID: 36839492

- 49. Garrett-Jones C, Shidrawi GR. Malaria vectorial capacity of a population of Anopheles gambiae: an exercise in epidemiological entomology. Bull World Health Organ. 1969; 40: 531–545. PMID: 5306719
- Liu J, Liu Y, Nie K, Du S, Qiu J, Pang X, et al. Flavivirus NS1 protein in infected host sera enhances viral acquisition by mosquitoes. Nat Microbiol. 2016; 1: 16087. https://doi.org/10.1038/nmicrobiol.2016.87 PMID: 27562253
- Liu Y, Liu J, Du S, Shan C, Nie K, Zhang R, et al. Evolutionary enhancement of Zika virus infectivity in Aedes aegypti mosquitoes. Nature. 2017; 545: 482–486. https://doi.org/10.1038/nature22365 PMID: 28514450
- 52. Kuo L, Jaeger AS, Banker EM, Bialosuknia SM, Mathias N, Payne AF, et al. Reversion to ancestral Zika virus NS1 residues increases competence of Aedes albopictus. PLoS Pathog. 2020; 16: e1008951. https://doi.org/10.1371/journal.ppat.1008951 PMID: 33052957
- **53.** Muller DA, Young PR. The flavivirus NS1 protein: molecular and structural biology, immunology, role in pathogenesis and application as a diagnostic biomarker. Antiviral Res. 2013; 98: 192–208. <u>https://doi.org/10.1016/j.antiviral.2013.03.008</u> PMID: 23523765
- 54. Alizon S, Hurford A, Mideo N, Van Baalen M. Virulence evolution and the trade-off hypothesis: history, current state of affairs and the future. J Evol Biol. 2009; 22: 245–259. <u>https://doi.org/10.1111/j.1420-9101.2008.01658.x PMID: 19196383</u>